

### Research Article

## An Automatic Image Processing Method Based on Artificial Intelligence for Locating the Key Boundary Points in the Central Serous Chorioretinopathy Lesion Area

# Jianguo Xu,<sup>1</sup> Jianxin Shen<sup>()</sup>,<sup>1</sup> Cheng Wan,<sup>2</sup> Zhipeng Yan,<sup>3</sup> Fen Zhou<sup>()</sup>,<sup>3</sup> Shaochong Zhang,<sup>4</sup> and Weihua Yang<sup>()</sup>

<sup>1</sup>College of Mechanical and Electrical Engineering, Nanjing University of Aeronautics and Astronautics, Nanjing 210016, China
 <sup>2</sup>College of Electronic and Information Engineering, Nanjing University of Aeronautics and Astronautics, Nanjing 210016, China
 <sup>3</sup>The Affiliated Eye Hospital of Nanjing Medical University, Nanjing 210029, China
 <sup>4</sup>Shenzhen Eye Hospital, Jinan University, Shenzhen 518040, China

Correspondence should be addressed to Jianxin Shen; cadatc@nuaa.edu.cn and Weihua Yang; benben0606@139.com

Received 7 September 2022; Revised 8 October 2022; Accepted 25 January 2023; Published 10 February 2023

Academic Editor: Mamoon Rashid

Copyright © 2023 Jianguo Xu et al. This is an open access article distributed under the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

Accurately and rapidly measuring the diameter of central serous chorioretinopathy (CSCR) lesion area is the key to judge the severity of CSCR and evaluate the efficacy of the corresponding treatments. Currently, the manual measurement scheme based on a single or a small number of optical coherence tomography (OCT) B-scan images encounters the dilemma of incredibility. Although manually measuring the diameters of all OCT B-scan images of a single patient can alleviate the previous issue, the situation of inefficiency will thus arise. Additionally, manual operation is subject to subjective factors of ophthalmologists, resulting in unrepeatable measurement results. Therefore, an automatic image processing method (i.e., a joint framework) based on artificial intelligence (AI) is innovatively proposed for locating the key boundary points of CSCR lesion area to assist the diameter measurement. Firstly, the initial location module (ILM) benefiting from multitask learning is properly adjusted and tentatively achieves the preliminary location of key boundary points. Secondly, the location task is formulated as a Markov decision process, aiming at further improving the location accuracy by utilizing the single agent reinforcement learning module (SARLM). Finally, the joint framework based on the ILM and SARLM is skillfully established, in which ILM provides an initial starting point for SARLM to narrow the active region of agent, and SARLM makes up for the defect of low generalization of ILM by virtue of the independent exploration ability of agent. Experiments reveal the AI-based method which joins the multitask learning, and single agent reinforcement learning paradigms enable agents to work in local region, alleviating the time-consuming problem of SARLM, performing location task in a global scope, and improving the location accuracy of ILM, thus reflecting its effectiveness and clinical application value in the task of rapidly and accurately measuring the diameter of CSCR lesions.

#### 1. Introduction

CSCR is a common fundus macular disease, which causes the visual object to be deformed, darkened, or become smaller and is one of the factors affecting human visual health. But its pathogenesis is still unknown in ophthalmology. In recent years, some scholars have put forward new theories on the pathogenesis of CSCR, such as the theory of choroidal dysfunction and the theory of retinal pigment epithelium dysfunction, which have explained the pathogenesis of CSCR to a certain extent and appropriately promoted human cognition of the fundus disease. This macular disease is mostly seen in young men aged 30 to 50 and is typically characterized by neurosensory retinal detachment (NRD, as shown in Figure 1) with or without pigment epithelium detachment (PED) [1, 2]. Although the vision of some patients may recover spontaneously within a few months without any intervention, it is still difficult for some patients to recover to normal vision without surgery or drugs in a short time. In general, the main active



FIGURE 1: The typical lesion characteristic of CSCR: NRD.

interventions for the treatment of CSCR are laser surgery and drugs. No matter which method is adopted, it is essential and critical to carry out effective quantitative monitoring of the CSCR lesion area, which lays a foundation for timely obtaining the disease information and then assisting ophthalmologists to more objectively evaluate the severity of this disease and the efficacy of the corresponding treatment plan and also provides a basis for better formulating the follow-up treatment scheme.

At the moment, the monitoring parameters of CSCR mainly include the central macular thickness (CMT), best corrected visual acuity (BCVA), maximum height, and diameter of CSCR lesion area. In addition, the CSCR lesion area is also an important parameter, and its direct segmentation and indirect detection methods have been carried out by many researchers [3–9]. A fully convolutional neural network was built for the automatic segmentation of subretinal fluid, and with the help of shrinking and expanding network structure, an average dice rate of 0.91 was obtained [3]. To deal with the large variations of the locations and shapes of CSCR lesion and the low contrast of Bruch membrane areas, Xue et al. [4] proposed a deep ensemble neural-like P system that integrated the strengths of deep convolutional neural networks and the spiking neural P system and achieved a maximum average dice rate of 0.97, which showed great potential in actual application. Wu et al. [5] presented a two-stage scheme consisting of detecting fluid-associated abnormalities by using thickness map prior and segmenting the subretinal fluid by using the fuzzy level set with a spatial smoothness and was beneficial for the automatic quantification of lesion area. Similar to [3], an end-to-end pipeline [6] inspired by the SegNet neural network was adopted for the identification and segmentation of CSCR fluid regions, which facilitated a more complete analysis of CSCR. Based on loosely coupled level sets, Novosel et al. [7] raised a locally-adaptive approach for the segmentation of the fluid and the interfaces between retinal layers, and a dice coefficient for fluid segmentation of 0.96 was acquired, which revealed a great potential in quantifying the CSCR lesion area. Moreover, Zhen et al. [8] tried to detect CSCR based on the deep learning architecture and color fundus images. However, this method cannot describe CSCR lesions in detail, so it is not conducive to the monitoring of the disease. A commendable segmentation model combining the U-Net and generative adversarial network was ingeniously constructed by Yoo et al. [9]. To the best of our knowledge, this framework was the first time to achieve the segmentation of CSCR lesions in the color fundus images by developing a cascaded network, which is of great

significance for quantitative monitoring of CSCR by virtue of conventional fundus image examination. In addition to the previous direct segmentation schemes, there are also some indirect detection methods [10, 11]. Syed et al. [10] constructed a support vector machine (SVM) classifierbased model for the automated diagnosis of CSCR. Specifically, they established a feature vector with a length of 8 based on retinal thickness and cyst space cavity to guide the classifier to learn proper weights for judging the disease category. A similar idea was also designed by Khalid et al. [11], where the difference was that 9 extracted features and more testing samples were adopted to train the classifier for making more accurate judgments on the type of retinal diseases. In these schemes, the feature descriptors of the CSCR lesion area are firstly established by applying the feature engineering technique, and then, the classifier is trained by using the feature vectors to construct the lesion detection model. Since such schemes require detailed digital description of the lesions, professional cognition of the characteristics of the lesions is crucial.

Besides, the fluid segmentation of other fundus diseases also provides a reference for the area quantification method of CSCR lesion [12-17]. To detect three-dimensional retinal fluid (i.e., symptomatic exudate-associated derangements), Xu et al. [12] developed a novel voxel classification-based approach using a layer-dependent stratified sampling strategy, and this approach performed well in dealing with the class imbalance issue. By combining the squeeze-andexcitation blocks and the U-shape network, Chen et al. [13] put forward a structure called SEUNet to segment fluid regions in the age-related macular degeneration (AMD) and supplied an effective for fluid segmentation. Based on graph shortest paths and neutrosophic transformation, a fullyautomated segmentation method was designed for the accurate segmenting of diabetic macular edema (DME) biomarkers so as to provide a quantitative measure for DME diagnosis [14]. Alsaih et al. [15] employed four wide-spread deep learning models for the segmentation of three retinal fluids in AMD and explored how the patch-based technique pushes the performance of deep learning-based models, which was conducive to the improvement of such scheme. Lu et al. [16] presented a deep learning-based method for segmenting multiclass retinal fluids. Different from the common deep learning schemes, this method introduced the random forest classifier in postprocessing to reduce the over segmentation problem in the independent network model. Hassan et al. [17] also constructed a deep learning-based segmentation network integrating the atrous spatial pyramid pooling module, residual module, and inception module to segment multiclass retinal fluids and brought a considerable gain in efficiency.

The previous direct segmentation pipelines or indirect detection methods undoubtedly enrich the research ideas of the automatic quantification scheme of the CSCR lesion area, which is of great significance for the precise treatment of this eye disease. Nevertheless, the tediousness of the pixellevel annotation task in deep learning-based segmentation method and its potential defects of insufficient generalization ability, the strong dependence of feature engineering on professional experience in classical machine learning-based detection way, and the low accuracy and weak adaptability in traditional image processing-based segmentation plan may restrict the wide application of the previous methods in the quantification task of CSCR lesion area to a certain extent. It has to be said that the lesion diameter measurement scheme [18] based on locating key boundary points does appropriately alleviate the previous situations, but the gradientbased correction module (GBCM) in this scheme relies on setting appropriate threshold parameters and is sensitive to the position of the starting point provided by ILM.

Considering the limitations and advantages of the previous methods, as well as the challenges of diameter measurement caused by the diameter differences of CSCR lesion areas in different frames (as shown in Figure 2), this paper constructs an automatic image processing method (i.e., a joint framework) based on artificial intelligence for rapidly and accurately measuring diameter of CSCR lesion area from the perspective of locating key boundary points in the CSCR lesion area. The details are as follows: (1) In the first step, the multitask learning-based ILM is appropriately adjusted and used for rapid location of key boundary points in the CSCR lesion area, laying the foundation for subsequent accuracy improvement. (2) In the second step, the location task is described as MDP, in which the single agent aims to explore and lock the key boundary points in the CSCR lesion area through continuous interaction with the image environment. (3) The joint framework based on ILM and SARLM is skillfully designed to make up for the defect of low generalization of ILM by employing the unique exploration ability of agent in SARLM and to narrow the active region of single agent by providing the initial starting point for SARLM through the ILM. (4) In the fourth step, extensive and in-depth experiments are carefully carried out to prove and analyze the effectiveness and feasibility of the joint framework in the key boundary point location task of CSCR lesion area and its application effect in the lesion diameter measurement.

The structure of the remaining part is as follows: Section 2 describes the related works of multitask learning and single agent reinforcement learning. Section 3 explains the implementation details of our proposed method. Section 4 shows the results and discussions. Section 5 concludes the research work.

#### 2. Materials and Related Methods

*2.1. Materials.* The CSCR source images used in the experiments are provided by the cooperative eye hospitals, and

the patients' privacy information has been desensitized carefully. The annotation task of all the CSCR B-scan images is jointly completed and reviewed by professional oph-thalmologists and relevant academic personnel. After the conventional data augmentation operations, the number of image and annotation pairs in the dataset used for training reaches 3240. Additionally, to evaluate the effect of the joint framework in the testing dataset, a total of 25 patient-level data, including 912 OCT B-scan images, are introduced into this process.

#### 2.2. Related Methods

2.2.1. Multi-Task Learning. As one of the artificial intelligence technologies, multitask learning [19] is a learning paradigm that improves generalization ability of the convolutional neural network model by using the domain information contained in the training signals of related tasks as an inductive bias, which has been extensively applied in downstream tasks such as object detection, target classification, and semantic segmentation [20-22]. Meanwhile, this paradigm also shines brightly in various medical image processing tasks [18, 23-25]. For assisting the diameter measurement of the CSCR lesion area, the multitask learning paradigm was introduced into the key boundary point location task for the first time [18], enabling the rapid locking of the relevant coordinates. To obtain a robust retinal disease grading model, Ju et al. [23] extracted additional monitoring signals from various sources by using multitask learning and achieved a significant improvement. A new canonical correlation analysis model [24] combining the biologically meaningful structures with the multitask learning framework was designed to mine the shared representations in multimodal data, which experimentally demonstrated the potential of multitask learning. Additionally, this paradigm also performed well in improving the accuracy of glaucoma diagnosis [25]. By sharing most of the parameters of the segmentation layers and classification layers, the feature representation ability of the model for a given task is enhanced, and then a win-win situation is achieved.

2.2.2. Reinforcement Learning. As a unique machine learning method to realize artificial intelligence, the reinforcement learning (RL) model has emerged in various scenes with its unique operating principle, in which the artificial agent obtains rewards and punishments through the continuous interaction with the environment [26] and then learns the optimal strategy for a given task. In particular, RL has shown satisfactory performance in various tasks in the field of medical image processing, such as registration, classification, and segmentation. In the registration task [27, 28], instead of directly optimizing an image matching metric, the goal of artificial agent was to find the best sequence of motion actions to achieve the best alignment between images. In the classification task [29], the agent cropped the appropriate patch on the original image through hard attention mechanism and updated the cropping strategy with the feedback of the classification network,



FIGURE 2: The diameter difference of CSCR lesion area in B-scan images under different frames.

so as to achieve better classification accuracy of breast cancer. In the segmentation task [30], the process of lymph node segmentation was completed by the interaction of two networks, where the decision network provided the target bounding box for the segmentation network, and the output of the segmentation network guided the policy network to make better strategies. Moreover, RL has also been applied and performed well in landmark detection [31–34]. Different from the traditional machine learning schemes, in this kind of application, the object appearance and parameter search strategy are unified into a framework, in which the behaviour strategy of agent and the effective object feature representation are jointly learned to better achieve the given task.

The previous research explored and confirmed the feasibility and effectiveness of the application of multitask learning and RL in the corresponding scenes and also promoted the inspiration of our research ideas in this paper. The specific details will be shown in the following sections.

#### 3. The Proposed Method

As shown in Figure 3, it is the overall flow chart of the scheme proposed in this paper, including image preprocessing module (IPM), ILM, and SARLM. Firstly, IPM is employed to provide datasets for the subsequent training and testing steps of ILM and SARLM. Secondly, the operation of independently training ILM and SAILM based on the training dataset is carefully performed. Then, the testing images are input into the trained ILM model to obtain the preliminary results of key boundary points. Finally, the testing images and the corresponding location results are sent to SARLM to get the final results.

3.1. Motivation. Through the previous brief analysis, it can be clearly found that both the multitask learning and reinforcement learning have achieved a wide layout in various visual tasks and obtained gratifying results. In the previous applications, the multitask learning paradigm does improve the adaptability of the deep learning model to a certain extent. Nevertheless, the paradigm usually works independently in the downstream tasks, thus resulting in the generalization of the multitask model that is still affected by factors such as the volume of data and network structure. Although the RL model performs well in different scenarios, agents usually regard the global region of the input image as the interactive environment, which is bound to lead to a significant increase in the time cost and computing power required to complete the task. This cannot help but bring some thoughts to our research in this paper, that is, whether these two learning paradigms can be integrated to alleviate the above issues. On the one hand, ILM is used to realize the preliminary and rapid location of key boundary points to provide the initial starting point for the RL model, which in turn achieves the reduction of the active region of artificial agent. On the other hand, based on the unique exploration ability of agent in the RL model, the position of key boundary points is further adjusted on the basis of the initial location results of ILM in a local range. The specific implementation route and experimental results will be detailed in the following sections. It is the successful application and surprising achievements of these two learning paradigms in various visual tasks that encourage us to make further attempts in this key boundary point location task.

3.2. The Preprocessing Step. Due to the equipment and human factors, the quality and size of medical images initially obtained from the clinic are often unable to directly adapt to the downstream tasks, so the image preprocessing operation is particularly critical. In this paper, the source images acquired from the clinic are in a whole composed of the scanning laser ophthalmoscope (SLO) part and the OCT B-scan part, which cannot be directly applied to the key boundary points location tasks. In view of this, we use the separation operation designed in our previous work [18] to realize the separation of the previous parts. In addition, considering that the image size and speckle noise may interfere with the performance of both the ILM and SARLM, the clipping operation and the BM3D (Block-matching and 3D filtering [35])-based denoising operation are then applied to OCT B-scan images. The overall process of IPM is shown in Figure 4. After the image preprocessing, the size and quality of OCT B-scan images have been improved and then followed by image annotating step which is completed



FIGURE 3: An overall flow chart.



FIGURE 4: The image preprocessing process.

by professional ophthalmologists. Finally, the OCT B-scan image dataset used to locate the key boundary points in the CSCR lesion area is established, which paves the way for the follow-up work.

#### 3.3. The Joint Framework

*3.3.1. ILM.* Inspired by the excellent performance of the multitask model in face key point detection [36], we adjusted

the architecture appropriately for the first time and introduced it into the key boundary point location scene in the CSCR lesion area [18], realizing another application test of this paradigm. In this paper, ILM continues to serve the task of initial location of key boundary points, and its specific composition is shown in Figure 5. The residual network [37] and MobileNet [38, 39] network are employed here as the CNN backbones to mine the background and nonbackground information contained in the OCT B-scan images, enabling the feature representation of CSCR lesion



FIGURE 5: The overall framework of ILM.

information. Specifically, we constructed the MobileNet with a width multiplier of 0.25 (i.e., MB1-0.25) [38] and its improved version (i.e., MB2-0.25) based on the inverted residual network [39], which are regarded as two kinds of backbone networks. Taking into account the network parameters and experimental conditions, resnet18 (i.e., R18), resnet34 (i.e., R34), and resnet50 (i.e., R50) are selected as another three kinds of backbone networks. The previous five backbone network structures are shown in Table 1. In addition, considering the capacity limitation in this paper, the FPN module [40], context module, and multitask loss module [41] will not be repeated further.

3.3.2. SARLM. As previously analyzed, RL has been popularized in various visual tasks; especially, its successful application in landmark detection promotes the proposal of our scheme. It should be noted that considering the distribution characteristics of key boundary points in this task and the time cost of agent interaction with the environment, this paper establishes SARLM to deal with the location task based on a single agent. The overall framework of SARLM is shown in Figure 6(a). Since the key boundary points are located on both sides of the CSCR lesion area, the SARLMs based on the left agent and the right agent are designed respectively. Although the structure of the two SARLMs is the same, the training process is carried out separately. Unlike the traditional machine learning scheme, the training samples required by SARLM are obtained through the continuous interaction between the agent and the environment, which are stored in the experience memory. The terms involved in the process are as follows:

(i) State: This term describes the surrounding environment including the location of the agent, which is mainly divided into the current state and the next state. In this task, in order to improve the operation efficiency of the agent, based on the initial location point provided by ILM, we first limit the active region of the agent to the purple square box (as shown in Figure 6(a)) with the size of 80. Then, with the location of the agent as the center, a square region with a size of 32 is cropped on the B-scan image as the state.

- (ii) Action: This term refers to the moving direction of the agent in the environment, which is used to realize the interaction between the agent and the environment. In this paper, we set up four discrete actions, namely, up, down, left, and right, to control the agent to move in the corresponding direction with a step of one pixel, so as to achieve its exploration of the environment.
- (iii) Reward: This term denotes the feedback of an agent after taking an action, aiming at evaluating whether the current action is conducive to the agent to achieve the given task. In the task of locating the key boundary points in the CSCR lesion area, the difference of the Euclidean distances between the agent and the target point before and after the action is regarded as reward. In addition, in order to avoid excessive Q value and obtain good conditional gradient, reward is clipped between -1 and 1 according to the common operation. The reward function is defined as follows:

$$R = D(P_i, P_T) - D(P_{i-1}, P_T).$$
(1)

- (iv) Policy: This term is a mechanism to determine the behavior of an agent. It is a mapping from the current state of the agent to the corresponding behavior taken by the agent. It defines various possible behaviors and corresponding probabilities of the agent in each state. In the key boundary point location task, the strategy is the behavior selection mechanism that enables the agent to reach the key boundary point of the CSCR lesion by a series of optimal actions. In the process of taking the optimal action, the agent can obtain the maximum cumulative reward.
- (v) Termination: This term is used to define the stopping rules of agents in the training or testing stages, so as to prevent the agents from exploring and exploiting in the environment indefinitely. In this paper, in the training stage, we define the termination flag as true when the Euclidean distance between the agent and the target point is less than or equal to one pixel.

| Layer name | Output size   | R18   | R34   | R50   | MB1-0.25  | MB2-0.25   |                                  |
|------------|---------------|---|---|---|---|--|----------------------------------|
| Convl      | C5: 160 × 160 |   | 7×7 64 stride 2   | 3×3, 8, stride 2<br>3×3, 8  | $3 \times 3$ , 8, stride 2<br>$1 \times 1$ , 48                                 |  |                                  |
|            | 00.100 / 100  |   | , , , , , , , , , , , , , , , , , , ,                                     | 1×1, 16   | 3×3, 48<br>1×1, 16  |  |                                  |
|            |               | 3   | ×3 max pool, stride   | 2   | 3×3, 16, stride 2   | 1×1, 96  |                                  |
| Conv2      | C4: 80×80     | $\begin{bmatrix} 3 \times 3, 64 \\ 3 \times 3, 64 \end{bmatrix} \times 2$       | $\begin{bmatrix} 3 \times 3, 64 \\ 3 \times 3, 64 \end{bmatrix} \times 3$ | $\begin{bmatrix} 1 \times 1, & 64 \\ 3 \times 3, & 64 \end{bmatrix} \times 3$   | $1 \times 1, 32$<br>$3 \times 3, 32$  | 3 × 3, 96, stride 2<br>1 × 1, 32   |                                  |
|            |               | $\begin{bmatrix} 3 \times 3, 64 \end{bmatrix}$                                  |   | $\lfloor 1 \times 1, 256 \rfloor$   | 1×1, 32   | $1 \times 1, 192, 5 \times 5, 192$<br>$1 \times 1, 32$   |                                  |
| Copy3      | C3: 40 × 40   | $\begin{bmatrix} 3 \times 3, & 128 \\ 3 \times 3, & 128 \end{bmatrix} \times 2$ | $\begin{bmatrix} 3 \times 3, 128 \\ 2 & 2 & 122 \end{bmatrix} \times 4$   | $\begin{bmatrix} 1 \times 1, & 128 \\ 3 \times 3, & 128 \end{bmatrix} \times 4$ | $3 \times 3$ , 32, stride 2<br>1 × 1, 64<br>3 × 3, 64                           | $1 \times 1, 192$<br>$3 \times 3, 192, stride 2$<br>$1 \times 1, 64$                           |                                  |
|            |               |   | $\begin{bmatrix} 3 \times 3, 128 \end{bmatrix}$                           | 1 × 1, 512  | 1×1, 64   | $1 \times 1$ , 384; $3 \times 3$ , 384<br>$1 \times 1$ , 64                                    |                                  |
| Conv4      | C2: 20 × 20   | $\begin{bmatrix} 3 \times 3, 256 \end{bmatrix} \times 2$                        | $\begin{bmatrix} 3 \times 3, 256 \end{bmatrix} \times 6$                  | $\begin{bmatrix} 1 \times 1, 256 \\ 3 \times 3, 256 \end{bmatrix} \times 6$     | 3×3, 64, stride 2<br>1×1, 128   | 1 × 1, 384<br>3 × 3, 384, stride 2<br>1 × 1, 128   |                                  |
| Conve      | 02. 20 × 20   | $\begin{bmatrix} 3 \times 3, 256 \end{bmatrix}^{2}$                             | $\begin{bmatrix} 3 \times 3, 256 \end{bmatrix}^{\times 0}$                | $\begin{bmatrix} 3 \times 3, & 250 \\ 1 \times 1, & 1024 \end{bmatrix}$         | $\begin{bmatrix} 3 \times 3, & 128 \\ 3 \times 3, & 128 \end{bmatrix} \times 5$ | $\begin{bmatrix} 1 \times 1, 768 \\ 3 \times 3, 768 \\ 1 \times 1, 128 \end{bmatrix} \times 5$ |                                  |
| Conv5      | C1: 10 × 10   | $\begin{bmatrix} 3 \times 3, 512 \end{bmatrix}_{\times 2}$                      | $\begin{bmatrix} 3 \times 3, 512 \end{bmatrix} \times 3$                  | $\begin{bmatrix} 1 \times 1, 512 \\ 3 \times 3, 512 \end{bmatrix} \times 3$     | 3×3, 128, stride 2<br>1×1, 256<br>3×3, 256                                      | $1 \times 1, 768$<br>$3 \times 3, 768, stride 2$<br>$1 \times 1, 256$                          |                                  |
|            | 01. 10 ~ 10   | 01. 10 × 10   | $\begin{bmatrix} 3 \times 3, 512 \end{bmatrix}^{\times 2}$                | $\begin{bmatrix} 3 \times 3, 512 \end{bmatrix}^{\times 5}$                      | $\begin{bmatrix} 1 \times 1, & 2048 \end{bmatrix}$                              | 1×1, 256   | 1×1, 1536; 3×3, 1536<br>1×1, 256 |

TABLE 1: The five backbone networks used in ILM.



FIGURE 6: The overall framework of SARLM and deep Q-network: (a) SARLM; (b) deep Q-network.



where  $N_{\text{Train}}$  denotes the maximum iteration value to limit the number of times the agent implements the target point location operation in the environment during training stage, and its value is empirically set to 100. In the testing stage, because the ground truth of the target point cannot be provided, we design the following termination rule according to the experimental observation:

$$\delta_{Q} = \operatorname{abs}\left(\frac{1}{n}\sum_{j=i-15}^{i-8}Q_{j} - \frac{1}{n}\sum_{k=i-7}^{i}Q_{k}\right),$$

$$T_{\operatorname{test}} = \begin{cases} \begin{cases} \operatorname{True, if } \delta_{Q} < Tr \text{ and } q = T_{q}, \\ \\ \operatorname{False, else,} \\ \\ \\ \operatorname{True,} \end{cases} & \text{if step} < N_{\operatorname{Test}}, \end{cases}$$

$$(3)$$

where  $\delta_Q$  represents the difference between the average value of the first eight elements in the last 16 Q values and the average value of the last eight elements in the last 16 Q values. Tr is the threshold which is set to 0.3. q is used to confirm whether the agent has converged to the target point, and its corresponding threshold  $T_q$  is set to 2 in this paper according to the time cost and experimental observation.  $N_{\text{Test}}$  denotes the maximum iteration value to limit the number of times the agent implements the target point location operation in the environment during testing stage, and its value is empirically set to 60.

After the previous brief introduction and analysis, it is followed by the typical reinforcement learning paradigm (i.e., Q-Learning) [42]. In various RL scenarios, appropriate actions are the core to achieve effective and continuous interaction between agent and environment, the optimization process of which can be completed based on the state-action value function Q(s, a) [43]. By solving the following formula, the corresponding Q value can be obtained after implementing the corresponding action in each state, and then the best action can be selected depending on the highest long-term Q value.

$$Q_{i+1}(s,a) = E\left[R + \gamma \max_{a'} Q_i(s',a')\right],$$
 (4)

where  $Q_{i+1}$  and  $Q_i$  represent the Q values at step I and i+1, respectively. s and s' denote the current state and the next state, respectively. Correspondingly, a and a' are the current action and the next action, respectively.  $\gamma$  is the discount factor and is set to 0.95 in this paper. However, when there are too many state-action pairs in Q-learning, the Q-Table will be too large and cause excessive memory consumption. At this time, the  $Q(s, a; \theta)$  obtained by approximating Q(s, a) with the help of the deep neural network (i.e., Deep Q-Network, DQN) can avoid the previous issue. By periodically assigning the weight parameters of the current Q-Network to the target Q-Network, the parameters of the Q-Network can be continuously updated. Specifically, the parameters of DQN can be obtained by solving the following equation:

$$L_{\text{DQN}}(\theta) = E\left[\left(R + \gamma \max_{a'} Q\left(s', a'; \theta'\right) - Q\left(s, a; \theta\right)\right)^2\right],\tag{5}$$

where  $\theta'$  and  $\theta$  are the parameters of target Q-Network and the current Q-Network, respectively. In this paper, the DQN structure is designed as shown in Figure 6(b). The input size of the network is  $32 \times 32 \times 4$ . During the training or testing stage, the input image is composed of four cropped single channel patches with the size of  $32 \times 32$ . After a series of convolutional layers, pooling layers, fully-connected layers, and various activation functions, a four-dimensional *Q*value vector can be obtained, and then the corresponding action can be selected for the agent to interact with the environment.

*3.3.3. ILM-SARLM.* On the basis of the previous research, the joint framework shown in Figure 7 is finally established, mainly consisting of three parts, namely, ILM part, SARLM part, and joint location part. The execution contents of each part are as follows:

- ① ILM is firstly trained based on all the OCT B-scan images in the training dataset. Since five CNN backbone networks are applied in the ILM framework, the training process needs to be repeated five times. Then, offline testing models corresponding to various backbone networks can be acquired.
- ② It should be pointed out that the location model of key boundary points on both sides of the CSCR lesion area is trained independently. Although the network structure on both sides is the same, the weight parameters will not be shared between them. Under the previous premise, SARLMs are trained based on some OCT B-scan images in the training dataset to obtain independent DQN models for the action selection of left and right agents, respectively.
- ③ On the basis of ① and ②, the location task of key boundary points is achieved through a cascade operation. Specifically, the testing image is firstly sent to the offline testing model to obtain the initial coordinates of the key boundary points on both sides. Then, the testing image and its corresponding initial location results are sent to the DQN models, and the agents further optimize the initial points of ILM again in the purple square active region (as shown in Figure 7) which is delimited according to the initial points on both sides.

3.4. The Backbone Networks and Algorithm Step. This section shows the structures of five CNN backbone networks used in ILM, and the specific details are shown in Table 1. In addition, the algorithm steps of ILM are detailed in [18], and the algorithm steps of SARLM and joint location parts are shown in Table 2.

#### 4. Experimental Settings

In order to verify the feasibility and effectiveness of the proposed scheme, this paper has conducted in-depth and extensive experiments. The training dataset, testing dataset, parameter settings and equipment conditions, and

#### Computational Intelligence and Neuroscience



FIGURE 7: The overall framework of ILM-SARLM.

TABLE 2: The algorithm of ILM-SARLM.

Input  $D_{Train}$ ,  $N_{Train}$ ,  $\gamma$ , learning rate, batch size *B*, epoch, max episode *P*, update frequency *L*, sample step *S*;

```
*Training step for SARLM**
//Initialize \theta, \theta', and Experience memory M;
for each K in epoch do
  //Shuffle D_{Train};
//While (j \le D_{Train}); select the j^{th} B-scan image B^j;
      //Initialize Experience memory M;
      for each k in Pdo
         Initialize T_{train} = False and step;
While notT_{train} and step < N_{Train};
               //Agent interact with environment E using a selected randomly
            or calculated by \operatorname{argmax}_{a}Q(s, a; \theta), and receive next state s';
               //Get reward R using (1) and termination flag T_{train} using (2);
         //Store (s, a, R, s', T_{train}) in M;
         if M \ge batch size and M \% S == 0;
            //Randomly select samples at size of B from M;
            \mathbf{if}T_{train};
               //Calculate L_{\text{DON}}(\theta) = E[(R - Q(s, a, \theta))^2];
               //Update \theta using Adm optimizer;
               //Update \theta' = \theta every L steps;
            else
               //Calculate (5);
               //Update \theta_{i} using Adm optimizer;
               //Update \theta' = \theta every L steps;
            end
         end
         // step++;
      end
      // j ++;
end
```

TABLE 2: Continued.

| Input | $D_{Train}$ , | N <sub>Train</sub> , | γ, | learning | rate, | batch | size E | 3, e | poch, 1 | max | episode | P, ι | ıpdate | frec | juency | y L, sa | nple ste | p S |
|-------|---------------|----------------------|----|----------|-------|-------|--------|------|---------|-----|---------|------|--------|------|--------|---------|----------|-----|
|-------|---------------|----------------------|----|----------|-------|-------|--------|------|---------|-----|---------|------|--------|------|--------|---------|----------|-----|

**Input**  $D_{Test}$ , Tr,  $N_{Test}$ ,  $T_q$ ; \*\*Testing step for ILM-SARLM\*\* //Initialize *q* and *step*; //Get the initial coordinates  $(x_L, y_L)$  and  $(x_R, y_R)$  using the offline testing ILM; //Send the initial results and testing images into the DQN models; While True; //Agents interact with environment E using actions calculated by  $\operatorname{argmax}_{a}Q(s, a; \theta)$ ; //Calculate  $\delta_{\Omega}$  using (3); *//step++;* if step <  $N_{Test}$  and  $\delta_Q$  < Tr; //q + = 1;end //Calculate  $T_{test}$  using (3) and (4); **if** $T_{test}$ **or**  $step = = N_{Test}$ ; //Agents stop interaction behaviour and output the final location coordinates  $(x_{\text{new}\_L}, y_{\text{new}\_L})$  and  $(x_{\text{new}\_R}, y_{\text{new}\_R})$ ; Break

evaluation metric involved in the experiments will be briefly described.

4.1. Parameter Settings and Equipment Conditions. Both ILM and SARLM are based on the TensorFlow framework as the development platform, and the computing power involved in the training and testing process of these models is mainly supplied by NVIDIA-3080ti GPU. For ILM, the parameters of epoch, learning rate, batch size, and optimizer are set to 40, 0.001, 20, and Adam, respectively; [18]can be referred for other settings. For SARLM, the parameters of epoch, learning rate, batch size, max episode, update frequency, sample step, and optimizer are set to 80, 0.0001, 32, 25, 50, 5, and Adam, respectively.

4.2. Evaluation Metric. In order to quantify the location accuracy of the key boundary points in the CSCR lesion area so as to evaluate the performance of the proposed scheme, this paper adopts an average Euclidean distance (AED) evaluation metric, which is expressed as follows:

$$AED = \frac{1}{2N_T} \sum_{i=1}^{N_T} (\|LG\|_2 + \|RG\|_2).$$
(6)

For left key point,

$$LG = (x_{\text{new}\_L} - x_{\text{truth}\_L}, y_{\text{new}\_L} - y_{\text{truth}\_L}).$$
(7)

For right key point,

$$RG = (x_{\text{new}_R} - x_{\text{truth}_R}, y_{\text{new}_R} - y_{\text{truth}_R}), \qquad (8)$$

where *LG* and *RG* are vectors determined by the coordinate of the left key point (i.e.,  $(x_{new\_L}, y_{new\_L}))$ , its corresponding ground truth value (i.e.,  $(x_{truth\_L}, y_{truth\_L}))$ , the coordinate of the right key point (i.e.,  $(x_{new\_R}, y_{new\_R}))$ , and its corresponding truth value (i.e.,  $(x_{truth\_R}, y_{truth\_R}))$ , respectively.  $\|\cdot\|_2$  represents the 2-norm to perform the calculation of Euclidean distance, and  $N_T$  denotes the number of OCT B-scan images included in a single patient data. The diameter measurement of the CSCR lesion area is also based on the 2-norm, and the specific equation is as follows:

$$D = \|LR\|_2,\tag{9}$$

$$LR = (x_{\text{new}\_R} - x_{\text{new}\_L}, y_{\text{new}\_R} - y_{\text{new}\_L}), \qquad (10)$$

where *LR* is the vector determined by the left and right key boundary points.

#### 5. Results and Discussions

This section analyzes and discusses the experiment. Before that, we will explain the terms involved in this process. It should be pointed out that in order to verify the effectiveness and feasibility of the joint framework proposed in this paper, not only the pure DQN based SARLM is combined with each kind of ILMs but also the DDQN-based [44, 45], Duel DQN (DuelDQN)-based [46], and Duel DDQN (DuelDDQN)-based SARLMs are introduced into the joint framework. As mentioned earlier, the SARLM model for locating the key boundary points on both sides in this paper is independent of each other, and the training process is carried out separately. In this way, the left key point location model based on DQN is named DQN\_LP, and the corresponding location model for the right key point is named DQN\_ RP. According to the same rule, DDQN\_LP, DDQN\_RP, DuelDQN\_LP, DuelDQN\_RP, DuelDDQN\_LP, and DuelDDQN\_RP can be obtained, respectively. Moreover, for different ILMs, we name it according to the name of the CNN backbone network in Section 3.3.1. We take the R18 backbone network and DQN for example, the pure ILM is named R18-Base, and the corresponding joint framework is called R18-DQN. The names of other joint frameworks also comply with this rule.

5.1. Convergence Observation of ILMs and SARLMs on the Training Dataset. Properly judging whether the model converges in the training process is an indispensable link related to the later model performance test and actual deployment. As shown in Figure 8(a), each ILM can converge from a large value to a small loss value after 6480 iterations, revealing that the weight parameters of the model can better fit a nonlinear function to deal with the task given in this paper. In addition, it can also be found that the progress trend of each task loss curve in ILM is almost the same, and the difference between losses is very small, which has important reference significance for setting the proportion parameter of each task item in the total loss. For SARLM, each model was trained with 80 epochs on the training dataset including 30 OCT B-scan images, and the total training time of eight models was about 240 hours, that is, 10 days. In order to boost the adaptability of SARLM to the initial position during the testing stage, the initial coordinate of the agent is initialized randomly in each episode of training, based on the coordinate ground truth of each key boundary point and the corresponding margin randomly selected from [-4, -6, -8, -10, -12, -14, 4, 6, 8, 10, 12, 14]. It can be clearly observed from Figure 8(b) that compared with the initial period, the reward value of each SARLM in the later training period finally stabilized within a certain range, which implies that through continuous exploration and exploitation, the agent gradually learned how to formulate appropriate behaviour strategies according to its environment to achieve the location task of the key boundary points in the CSCR lesion area. The previous observations show that the feasibility of ILMs and SARLMs in the training dataset has been preliminarily verified, laying a foundation for subsequent analysis.

5.2. Performance Analysis of SARLMs on the Validation Dataset. In order to select the appropriate SARLM for later application in the testing dataset, we conducted relevant experiments on the validation dataset consisting of 10 OCT B-scan images. Specifically, in order to check the location performance of SARLM under the random starting point in the local region of CSCR lesion, we designed 16 random initialization starting points for both the left and right agents based on the coordinate ground truth of the key boundary points and their corresponding margins in the X-axis direction and Y-axis direction (i.e., [-4, -14, 4, 14] for the Xaxis direction and [-4, -14, 4, 14] for the Y-axis direction). Under the previous settings, the AED curves of various SARLMs based on 80 epochs are obtained as shown in Figure 9. On the whole, in the later stage of training, the AED value of each SARLM on the validation dataset decreases compared with the initial stage, which indicates that the ability of the agent to locate the key point under the condition of random initial position is appropriately improved with the increase of learning times, corresponding to the hint of the reward convergence curve of SARLM on the training dataset. Moreover, the overall trend of the AED curves of SARLMs for the left and right key points in the same environment is similar, indicating that even if the local regions

of the key points on both sides are different, the strategy of training the SARLMs on both sides independently in this paper can make the agents properly adapt to this change.

Q value is an important indicator to measure the closeness of agents to target points and is the key basis to terminate the interaction between agents and the environment. As pointed out in [29], when the agent approaches the target point, Q value is relatively small; otherwise, it is relatively large. This view is further confirmed by the experimental results involved in the task of this paper. As shown in Figure 10, each best SARLM-selected based on the minimum AED can almost converge from the larger Q value at the initial time to the smaller Q value at the final stage, which means that the agent can perform well in the validation dataset after continuous learning. In some cases, when the initial position of the agent is set to approach the target point, the Q value does not change much before and after convergence, suggesting that the agent judges that it is close to the target point fall into the local optimized point in advance. This situation is also where we strive to break in the future.

However, we can be gratified that when the initial position of the agent is far away from the target point, each kind of SARLM can stabilize at a small Q value after the iteration stops, which shows that even if the maximum distance between the agent and the target point is about 19 pixels in the local region, it can finally converge near the target point. In fact, the initial location accuracy of ILM is usually less than this value (as shown in Table 3), which proves that it is feasible to use ILM to narrow the active region of the agent for SARLM. Corresponding to Figures 10 and 11, it shows the visual performance of each SARLM in the key boundary point location task under 16 initial positions, in which the silver dotted line represents the final position of the agent corresponding to the initial position. What can be clearly captured is that the agent can finally converge near the target point at different initial positions. Although there are great differences in each initial position, each final position of the agent is very close, and in some cases, the agent even locks the same final point (such as Figure 11(a) DQN\_RP, DuelDDQN\_LP, DuleDDQN\_RP and so on). The previous analysis not only shows the good location ability of each SARLM on the validation dataset but also implies its low sensitivity to the initial position in the local region, which plays an important role in promoting the proposal of the joint framework.

5.3. Performance of ILMs and ILM-SARLMs on the Testing Dataset. Based on the previous results and analysis, this section formally investigates the performance of the joint framework constructed based on the multitask learning and the single-agent reinforcement learning on the testing dataset, including qualitative analysis, quantitative analysis, and efficiency analysis, so as to explore and discover the value and potential of the framework in practical application scenarios and the corresponding details to be further improved.



FIGURE 8: Convergence curves of ILM and SARLM in the training process: (a) the loss curves of different ILMs [16]; (b) the reward curves of different SARLMs.

5.3.1. Qualitative Analysis. As shown in Figures 12 and 13, the Q value convergence and practical application effect of the joint framework composed of different ILMs and SARLMs under the key boundary point location task at the image level are shown, respectively. The icon on the right in Figure 12 represents the scanning number of all B-scan images in the patient-level data. In general, each joint framework can achieve convergence under a certain number of iterations, that is, obtain a smaller Q value in the later stage. However, similar to the observations in the validation dataset, the difference between the Q value of the agent before and after the key point location task in some B-scan

images of the testing dataset is small. This reveals that the initial starting point provided by ILM for SARLM is relatively close to the target point, thus making the agent wander around the target point all the time, and the external representation is the periodic small amplitude fluctuation of Q value. As shown in Figure 13, the reason behind the previous phenomenon is further demonstrated by the actual location effect. It can be clearly captured that sometimes the initial starting point is indeed in the region near the target point, resulting in the agent naturally interacting with the environment in a small range and gaining a small Q value, which echoes the observation in Figure 12. In addition, the cyan



FIGURE 9: The AED of SARLMs under different epochs.



FIGURE 10: Continued.



FIGURE 10: Q value convergence curve of SARLMs under different initial starting points: (a) example-1; (b) example-2; (c) example-3.

line in the figure represents the distance between the final location point of the joint framework and the target point, while the silver dotted line here denotes the distance between the starting point provided by ILM and the target point. It can be seen that in most cases, the length of the cyan line is smaller than that of the silver dotted line, which indicates that the joint framework has the ability to further optimize the position of the initial key boundary points. Meanwhile, Figure 13 also visually shows the performance difference between joint frameworks and conveys the level of various ILMs in the initial location task, which is conducive to the selection and deployment of the joint framework in practical application.

In addition to the previous qualitative analysis at the picture level, we also verify the ability of the different joint framework and the corresponding ILMs on patient-level

data, so the discussion based on patient level is also introduced here. Figure 14(a) clearly reveals the AED of each ILM and its corresponding joint framework on each patient-level data. It is obvious that compared with the pure ILMs, all kinds of joint frameworks can basically obtain lower AED values, further enriching the evidence of the effectiveness of the proposed model. Furthermore, the robustness of various joint frameworks and their ILMs at the patient level has also been carefully considered. As shown in Figure 14(b), compared with ILM, the box height of its corresponding joint framework is relatively small, showing that the performance of the joint framework on patient-level data fluctuates little, implying that the joint framework has good robustness. The cyan triangle in the figure denotes the average AED value of ILM or joint framework. The lower value also proves that the

|             | Model             | AED (pixels) | Size (M) |  |  |
|-------------|-------------------|--------------|----------|--|--|
|             | R18-DQN           | 3.71         |          |  |  |
|             | R18-DDQN          | 3.78         | 166.2    |  |  |
|             | R18-DuelDQN       | 3.61         | 100.2    |  |  |
|             | R18-DuelDDQN      | 3.91         |          |  |  |
|             | R34-DQN           | 3.71         |          |  |  |
|             | R34-DDQN          | 3.88         | 202.2    |  |  |
|             | R34-DuelDQN       | 3.72         | 283.2    |  |  |
|             | R34-DuelDDQN      | 3.95         |          |  |  |
|             | R50-DQN           | 3.79         |          |  |  |
| U.M. CADIM- | R50-DDQN          | 3.91         | 21.2.2   |  |  |
| ILM-SARLINS | R50-DuelDQN       | 3.72         | 313.2    |  |  |
|             | R50-DuelDDQN      | 4.06         |          |  |  |
|             | MB1-0.25-DQN      | 4.61         |          |  |  |
|             | MB1-0.25-DDQN     | 4.87         | 41       |  |  |
|             | MB1-0.25-DuelDQN  | 4.51         | 41       |  |  |
|             | MB1-0.25-DuelDDQN | 4.97         |          |  |  |
|             | MB2-0.25-DQN      | 3.68         |          |  |  |
|             | MB2-0.25-DDQN     | 3.78         | (())     |  |  |
|             | MB2-0.25-DuelDQN  | 3.62         | 66.3     |  |  |
|             | MB2-0.25-DuelDDQN | 4.01         |          |  |  |
|             | R18               | 5.73         | 138      |  |  |
|             | R34               | 8.21         | 255      |  |  |
| ILMs        | R50               | 6.21         | 285      |  |  |
|             | MB1-0.25          | 10.19        | 12.8     |  |  |
|             | MB2-0.25          | 5.40         | 38.1     |  |  |

TABLE 3: Comparison of ILMs and ILM-SARLMs based on the AED.



FIGURE 11: Continued.



FIGURE 11: The location of key boundary points of SARLMs under different initial starting points: (a) example-1; (b) example-2; (c) example-3.

joint framework possesses a more prominent key point location ability as a whole. What can also be seen is that there is a significant difference in robustness between joint frameworks, which, together with the information provided in Figure 13, sets a reference for the selection of models.

At the end of this section, the DQN-based SARLM is taken as an example to show the actual results of the joint frameworks under different ILMs in locating the key boundary points of CSCR lesion. As shown in Figure 15, with the ground truth as a reference, it is easily judged that various joint frameworks can achieve the effective reoptimization of the position of key boundary points on the basis of the initial results of ILM, which is consistent with the previous qualitative analysis based on the AED metric. Through the actual location results, the effectiveness and feasibility of the joint framework are demonstrated again.

5.3.2. Qualitative Analysis. This section further analyzes the performance between the joint framework and the corresponding ILM from the quantitative perspective. When the AED between the key boundary points on both sides provided by the joint framework and the corresponding target points is less than that between the key boundary points on both sides provided by the ILM and the corresponding target points, the joint framework successfully corrects the key boundary points on the B-scan image. As shown in Figure 16, from the image level, the successful correction rate of the key boundary point of the joint framework exceeds half of the total number of testing images, and the maximum



FIGURE 12: Continued.



FIGURE 12: Q value convergence curve of SARLMs under different ILMs: (a) MB1-0.25 (b) MB2-0.25 (c) R18 (d) R34 (e) R50.

value reaches 92.11% (i.e., a total of 840 B-scan images). Besides, there are some differences in the successful correction rate between joint frameworks based on five kinds of ILMs. However, in view of the defects of the weak generalization ability of ILM itself and the performance differences between them, it cannot be directly said that a smaller value corresponds to a joint framework with weak correction ability, and a larger value represents a strong ability. The successful correction rate here mainly reflects that the joint framework can further optimize the location of key boundary points based on the corresponding ILM. As for the comparison of performance differences between models, it can be based on the qualitative analysis in the previous section and the AED metric at the patient level shown in Table 3.

As shown in Table 3, the patient-level-based AEDs are counted. It can be clearly observed that each joint framework is significantly superior to the corresponding ILM in terms of this metric, obtaining an AED value with a minimum of 3.61 pixels, and the maximum difference between the two types of models is 5.68 pixels, which quantitatively shows the advantages of the joint framework. Additionally, due to the introduction of SARLM, the size of the joint framework is larger than that of the corresponding ILM, but this margin has little impact on the actual deployment. In general, the previous quantitative analysis also confirms the positive role of the joint framework in the given task.

5.3.3. Efficiency Analysis. This section focuses on analyzing the efficiency of the proposed joint framework and ILM. The time cost is the key reference to judge the efficiency of the model in completing the given task. In view of this, we recorded the time cost of ILMs and the corresponding joint frameworks in the key boundary point location task in the CSCR lesion. Figure 17(a) shows the time consumption



FIGURE 13: Continued.



FIGURE 13: Continued.



FIGURE 13: The location of key boundary points of SARLMs under different ILMs: (a) MB1-0.25; (b) MB2-0.25; (c) R18; (d) R34; (e) R50.







FIGURE 14: The performance of ILM-SARLMs under patient level: (a) the AED difference between models; (b) the robustness difference between models.



FIGURE 15: The display of location results of SARLMs under different ILMs.



FIGURE 16: The successful correction ratio of key boundary points of ILM-SARLMs under image level.



FIGURE 17: Comparison of time consuming between ILMs and ILM-SARLMs: (a) average time consuming at patient level; (b) average time consuming at image level.

based on the patient level, in which the time of ILM is significantly lower than that of the joint framework, and its minimum time consumption is only about 6.05 seconds. Furthermore, the time consumption based on the image level reflected in Figure 17(b) also displays the efficiency advantage of various ILMs. This situation is mainly due to the fact that SARLM in the joint framework needs to rely on the continuous interaction between the agent and the environment. Such an iterative process will naturally lead to an increase in time cost. To alleviate the previous issue, this paper aims to compress the time expenditure for the joint framework from two perspectives. On the one hand, the active region of the agent is limited to a local scope through the initial key boundary point location function of ILM, so as to avoid the transition time consumption caused by the global activities of the agent. On the other hand, the termination rule is properly developed to stop the repeated interaction behaviour of the agent in the later stage as soon as possible when it has approached the target point. The former can undoubtedly improve the efficiency of the agent. As for the latter, although the termination rule based on the Q value and the maximum iteration designed in this paper can appropriately stop the repeated wandering behaviour of the agent on time, better termination hints still need to be explored, which is also the perspective we need to further improve in the future.

5.4. Discussion on the Effectiveness of SARLMs for ILM-GBCMs on the Testing Dataset. The analysis and discussion in the previous sections show the superiority of the joint framework proposed in this paper over ILM. This section intends to explore whether SARLM can be equally effective in further improving the performance of ILM-GBCM [18] through qualitative and quantitative experiments. As shown in Figure 18, MB1-0.25-T50-P11, MB2-0.25-T40-P9, R18-T30-P9, R34-T20-P9, and R50-T20-P9 represent different ILM-GBCMs, respectively. Similarly, taking the DQN-based SARLM as an example, the joint frameworks based on the previous GBCMs are MB1-0.25-T50-P11-DQN, MB2-0.25-T40-P9-DQN, R18-T30-P9-DQN, R34-T20-P9-DQN, and R50-T20-P9-DQN, respectively. Figure 18(a) shows that the AED values of the joint frameworks at the patient level are higher than those of the corresponding ILM-GBCMs most of the time, indicating that the joint framework can further optimize the position of the key boundary points after the GBCM correction. It can also be found from Figure 18(b) that whether considering the average value of AED (the cyan triangle in the figure) or based on the box height, all kinds of joint frameworks are significantly better than their corresponding ILM-GBCMs, not only confirming the role of the joint framework in improving the location accuracy of key boundary points of ILM-GBCM but also revealing its good robustness in this task.

Furthermore, quantitative analysis experiments were carried out accordingly. As shown in Figure 19, the maximum successful correction rate of the joint framework is 83.00%, which shows that it has the ability to further improve the location accuracy on the basis of the correction results. However, by observing Figures 16 and 19, it can be found that compared with the successful correction rate of the location results of pure ILM, the improvement degree of SARLM on this metric of ILM-GBCM is relatively low, which stems from the fact that GBCM has played a corrective role in the initial location results of ILM to a certain extent, thus making the space for SARLM to further optimize the location accuracy of key boundary points smaller. Concurrently, AEDs of the two models based on the patient level are also emphatically recorded, as shown in Table 4.

Compared with ILM-GBCM, the AED values of the corresponding ILM-GBCM-SARLM are reduced, and the maximum difference between them is 3.64 pixels, which together with the image-level-based successful correction rate shows that SARLM also has a certain effect in boosting the ability of ILM-GBCM to locate the key boundary points in the CSCR lesion area.

5.5. Preliminary Application. Extensive experiments and analysis have proved the superiority of the proposed joint framework compared with the corresponding ILM in the key boundary point location task in the CSCR lesion area, which is embodied in the fact that it can make up for the weak generalization of ILM in this scene through the unique autonomous learning ability of agent in the lesion environment, which paves the way for its preliminary application in the actual measurement of the diameter of CSCR lesions in this section. Based on formulas (9) and (10) and the coordinates of the key boundary points output by the joint framework, the diameters of the CSCR lesions at all scanning angles can be measured quickly. As shown in Figure 20, the diameter measurement results of four patient-level samples, in which Figures 20(a) and 20(b) correspond to samples containing 24 OCT B-scan images, and Figures 20(c) and 20(d) correspond to samples containing 48 OCT B-scan images. On this basis, it is not only convenient for ophthalmologists to review the diameter size of the lesion at all scanning angles but also convenient for them to check the diameter size of the lesion at a certain scanning angle. Moreover, based on the diameter measurement results of lesions in all B-scan images, the maximum, minimum, and average values can also be obtained, providing a quantitative reference for ophthalmologists to judge the severity of the CSCR and evaluate the efficacy of the corresponding treatment scheme.

5.6. The Clinical Advantages of This Study in Ophthalmology. By and large, in the previous sections, both the qualitativequantitative analysis oriented to the evaluation of location effect and the time-cost consideration oriented to the evaluation of efficiency provide strong support for demonstrating the feasibility, effectiveness, and potential clinical application value of the proposed joint framework based on ILM and SARLM in the task of locating key boundary points in the CSCR lesion area. In addition, the qualitativequantitative experiments of the joint framework composed of SARLM and ILM-GBCM reveal that the introduction of SARLM can also add luster to the further optimization of the initial key boundary points provided by ILM-GBCM. On the basis of the previous part, based on the preliminary application of the diameter measurement of the CSCR lesion area, the potential deployment value of the proposed joint framework is affirmed from a practical point of view, and the significance of this study is further consolidated. It should be noted that although the scheme proposed in this paper is aimed at the diameter measurement of the CSCR lesion area, this method is also instructive for the design of the automatic measurement scheme of the diameter of the focus area in

#### Computational Intelligence and Neuroscience



FIGURE 18: The performance of ILM-GBCM-SARLMs under patient level: (a) the AED difference between models; (b) the robustness difference between models.

OCT B-scan images of other fundus diseases (such as diabetic macular edema, macular hole, and retinal angiomatous proliferation), which has potential guiding significance for more comprehensive monitoring of the lesion morphology and assisting ophthalmologists in the more objective assessment of patients' eye conditions.

5.7. Limitation Analysis and Future Work. Although this study reveals good clinical application value, the inadequacy of the proposed scheme still needs to be properly faced. Specifically, the joint framework may encounter the phenomenon of an agent wandering around the target point repeatedly in the procedure of processing a given task, which is also the factor leading to its slightly lower efficiency. We hold that this phenomenon may be due to the fact that the surrounding environment is too similar when the agent is approaching the target point in the process of interaction with the environment, resulting in the agent being unable to learn more effective behavior strategies and then falling into a local area and unable to extricate itself. For this problem, in the future, we plan to



FIGURE 19: The successful correction ratio of key boundary points of ILM-GBCM-SARLMs under image level.

|                     | Model                     | AED (pixels) | Size (M) |  |  |
|---------------------|---------------------------|--------------|----------|--|--|
|                     | R18-T30-P9-DQN            | 3.79         |          |  |  |
|                     | R18-T30-P9-DDQN           | 3.82         | 1// 0    |  |  |
|                     | R18-T30-P9-DuelDQN        | 3.61         | 166.2    |  |  |
|                     | R18-T30-P9-DuelDDQN       | 3.98         |          |  |  |
|                     | R34-T20-P9-DQN            | 3.63         |          |  |  |
|                     | R34-T20-P9-DDQN           | 3.74         | 202.2    |  |  |
|                     | R34-T20-P9-DuelDQN        | 3.54         | 283.2    |  |  |
|                     | R34-T20-P9-DuelDDQN       | 3.93         |          |  |  |
|                     | R50-T20-P9-DQN            | 3.78         |          |  |  |
| II M CDCM SADI Ma   | R50-T20-P9-DDQN           | 3.81         | 212.2    |  |  |
| ILWI-GDCMI-SARLINIS | R50-T20-P9-DuelDQN        | 3.56         | 515.2    |  |  |
|                     | R50-T20-P9-DuelDDQN       | 3.99         |          |  |  |
|                     | MB1-0.25-T50-P11-DQN      | 4.38         |          |  |  |
|                     | MB1-0.25-T50-P11-DDQN     | 4.50         | 41       |  |  |
|                     | MB1-0.25-T50-P11-DuelDQN  | 4.21         | 41       |  |  |
|                     | MB1-0.25-T50-P11-DuelDDQN | 4.64         |          |  |  |
|                     | MB2-0.25-T40-P9-DQN       | 3.79         |          |  |  |
|                     | MB2-0.25-T40-P9-DDQN      | 3.91         | (( )     |  |  |
|                     | MB2-0.25-T40-P9-DuelDQN   | 3.66         | 00.3     |  |  |
|                     | MB2-0.25-T40-P9-DuelDDQN  | 4.13         |          |  |  |
|                     | R18-T30-P9                | 4.79         | 138      |  |  |
|                     | R34-T20-P9                | 5.57         | 255      |  |  |
| ILM-GBCMs           | R50-T20-P9                | 4.65         | 285      |  |  |
|                     | MB1-0.25-T50-P11          | 7.85         | 12.8     |  |  |
|                     | MB2-0.25-T40-P9           | 5.21         | 38.1     |  |  |

TABLE 4: Comparison of ILM-GBCMs and ILM-GBCM-SARLMs based on the AED.

#### Computational Intelligence and Neuroscience

( http://

| B-scm 7                   | B-scan 8  | B-scan 9   | B-scan 10             | B-scan 11         | B-scan 12  |                   |
|---------------------------|---|------------|-----------------------|-------------------|--|-------------------|
|                           |   |            |                       |                   | 214  |                   |
| B-scan 13                 | B-scan 14   | B-scan 15  | B-scan 16             | B-scan 17         | B-scan 18  |                   |
|                           |   |            |                       |                   |  |                   |
| B-scan 19                 | B-scan 20   | B-scan 21  | B-scan 22             | H-scan 23         | B-scan 24  |                   |
| Max:337 Min:287 Aver:302  |   |            |                       |                   |  | 1000              |
|                           |   | (a)        |                       |                   | A CARE OF STREET, STRE | 10.00             |
| B-scan 1                  | B-scan 2  | B-scan 3   | B-scan 4              | B-scan 5          | B-scan 6   | N. 19 1. 19       |
|                           |   |            |                       |                   |  | -                 |
| B-scan 7                  | B-scan 8  | B-scan 9   | B-scan 10             | B-scan 11         | B-scan 12  | and ( d) ( m ) by |
|                           |   |            |                       |                   |  |                   |
| P man 13                  | B-scan 14   | B-scan 15  | B-scan 16             | B-scan 17         | B-scan 18  | 2.201.400         |
|                           |   |            |                       |                   |  | 1                 |
| B-scan 19                 | B-scan 20   | B-scan 21  | B-scan 22             | B-scan 23         | B-scan 24  |                   |
| M-16 Min 121 Arrs 140     |   |            | 16                    |                   |  |                   |
| Max:100 Mill:131 Aver:148 |   |            |                       |                   | a manage of the second   | 1.0.1 m           |
|                           |   | (b)        |                       |                   |  |                   |
|                           | B-scan 1  | B-scin 2   | B-scan 3 B-s          | ran 4 B-scan 5    | B-sem 6  | -                 |
|                           | B-scan 7  | B-scan 8   | B-scan 9 B-s          | can 10 B-scan 11  | H-scan 12  | and classes       |
|                           |   |            |                       |                   |  |                   |
|                           | B-scan 13   | B-scan 14  | B-scan 15 B-s         | can 16 B-syan 17  | B-scan 18  |                   |
|                           | B-scan 19   | B-scan 20  | B-scan 21 B-s         | can 22 B-scan 23  | H-scan 24  | to an             |
|                           |   |            |                       |                   |  |                   |
|                           | H-sean 25   | B-scan 26  | B-scan 27 B-s         | cun 28 B-scan 29  | B-scan 30  |                   |
|                           | B.con 31  | pri ny     | B-scan 33             | De Maria          |  |                   |
|                           |   |            |                       |                   |  |                   |
|                           | B-scan 37   | B-scan 38  | B-scan 39 B-          | can 40 B-scan 41  | B-scan 42  |                   |
|                           | B-scan 43   | B-scan 44  | B-scan 45 B-          | scan 46 B-scan 47 | B-scan 48  | -                 |
| ALANTITI TANA             |   |            |                       |                   |  | Second P          |
| Max:224 Min:200 Aver:210  |   |            |                       |                   |  |                   |
|                           |   | (c)        |                       |                   |  |                   |
|                           | B-scan I  | Baren 2    | B-scan 3              | B-sean 4 B-se     | n 5 B-scan 6   | -                 |
|                           | H-scan 7  | B-scan 8   | B-scan 9              | B-scan 10 B-sca   | m 11 B-scen 12   |                   |
|                           | 11/   |            |                       |                   |  |                   |
|                           | H-scan 13   | B-scan 14  | B-scan 15             | B-scan 16 B-sca   | n 17 B-scan 18   |                   |
|                           |   |            |                       | - these           |  | -                 |
|                           | B-scan 19   | B-scan 20  | B-scan 21             | B-scan 22 B-sca   | n 23 B-scar 24   | -                 |
|                           | H-sean 25   | B-scan 26  | B-scan 27             | B-scan 28 B-sc    | in 29 B-senn 30  |                   |
|                           |   | -          |                       |                   |  | 14                |
|                           | H-scan 31   | B-scan 32  | B-scan 33             | B-scan 34 B-sc    | in 35 B-scan 36  | _                 |
|                           |   | Direct 20  | Risson 39             | Descendo Desc     |  |                   |
|                           |   | D-solid 30 |                       |                   | Tresten 42   | -                 |
|                           | A second s |            | and the second second |                   |  |                   |

B-scan 3

B-scan 4

B-scan

B-scan 6

al luis

B-scan 2

B-scan 1

(d)

FIGURE 20: The diameter measurement of CSCR lesion area based on ILM-SARLM: (a) example-1; (b) example-2; (c) example-3; (d) example-4.

continue to mine solutions from the design of the termination rule and the adjustment of the DQN framework structure, so as to further improve the performance of the joint framework.

#### 6. Conclusions

An automatic image processing method (i.e., the joint framework) based on the multitask learning and single-agent reinforcement learning paradigms is constructed to achieve the goal of rapid and accurate location of key boundary points in CSCR lesion area, so as to facilitate the automatic diameter measurement of CSCR lesion. On the one hand, the adjustment and introduction of ILM initially realize the rapid locking of key boundary points and effectively narrow the activity range of agents, which helps to improve the location efficiency of SARLM. On the other hand, the unique exploration ability of the agent enables it to independently learn task-oriented behavior strategies, so that SARLM can better adapt to the differences of CSCR lesion areas in different scanning frames and appropriately make up for the defect of low generalization of ILM. Extensive experiments have been carried out carefully, demonstrating the effectiveness and feasibility of the joint framework in improving the location performance of ILM. The preliminary test on the diameter measurement of the CSCR lesion further reveals the potential clinical application value of the proposed joint framework, which also has a certain reference significance for the designation of the diameter measurement scheme of lesions in other fundus diseases (diabetic macular edema, macular hole, and retinal angiomatous proliferation). Generally speaking, the method proposed in this paper is a further innovation based on our previous work from the perspective of the algorithm, and in the future, we will pay attention to the inadequacy of this scheme and improve it.

#### **Data Availability**

The CSCR dataset used and analyzed in our research is available from the corresponding authors upon reasonable request.

#### **Conflicts of Interest**

The authors declare that they have no conflicts of interest regarding this paper.

#### Acknowledgments

This work was financially supported by Fundamental Research Funds for the Central Universities (No. NP2020420), China Postdoctoral Science Foundation Funded Project (No. 2019M661832), Shenzhen Fund for Guangdong Provincial High-level Clinical Key Specialties (SZGSP014), and Sanming Project of Medicine in Shenzhen (SZSM202011015). The authors would like to thank the cooperation units for providing the CSCR dataset for this scientific research

#### References

[1] A. Daruich, A. Matet, A. Dirani et al., "Central serous chorioretinopathy: recent findings and new physiopathology hypothesis," *Progress in Retinal and Eye Research*, vol. 48, pp. 82–118, 2015.

- [2] L. Yang, J. B. Jonas, and W. Wei, "Optical coherence tomography assisted enhanced depth imaging of central serous chorioretinopathy," *Investigative Opthalmology and Visual Science*, vol. 54, pp. 4659–4665, 2013.
- [3] T. Rao, G. Girish, A. Kothari, and J. Rajan, "Deep learning based sub-retinal fluid segmentation in central serous chorioretinopathy optical coherence tomography scans," in *Proceedings of the In the Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC)*, pp. 978–981, Berlin, Germany, July 2019.
- [4] J. Xue, Z. Wang, D. Kong et al., "Deep ensemble neural-like P systems for segmentation of central serous chorioretinopathy lesion," *Information Fusion*, vol. 65, pp. 84–94, 2021.
- [5] M. Wu, Q. Chen, X. He et al., "Automatic subretinal fluid segmentation of retinal SD-OCT images with neurosensory retinal detachment guided by enface fundus imaging," *IEEE Transactions on Biomedical Engineering*, vol. 65, no. 1, pp. 87–95, 2018.
- [6] J. Moura, J. Novo, M. Ortega, and M. Penedo, "Automated segmentation of the central serous chorioretinopathy fluid regions using optical coherence tomography scans," in *Proceedings of the In the IEEE 34th International Symposium on Computer-Based Medical Systems (CBMS)*, pp. 1–6, Aveiro, Portugal, June 2021.
- [7] J. Novosel, Z. Wang, H. Jong, M. Velthoven, K. Vermeer, and L. Vliet, "Locally-adaptive loosely-coupled level sets for retinal layer and fluid segmentation in subjects with central serous retinopathy," in *Proceedings of the In the IEEE 13th International Symposium on Biomedical Imaging (ISBI)*, pp. 702– 705, Prague, Czech Republic, Europe, April 2016.
- [8] Y. Zhen, H. Chen, X. Zhang et al., "Assessment of central serous chorioretinopathy depicted on color fundus photographs using deep learning," *Retina*, vol. 40, no. 8, pp. 1558–1564, 2020.
- [9] T. K. Yoo, B. Y. Kim, H. K. Jeong, H. K. Kim, D. Yang, and I. H. Ryu, "Simple code implementation for deep learningbased segmentation to evaluate central serous chorioretinopathy in fundus photography," *Translational Vision Science and Technology*, vol. 11, no. 2, p. 22, 2022.
- [10] A. M. Syed, T. Hassan, M. U. Akram, S. Naz, and S. Khalid, "Automated diagnosis of macular edema and central serous retinopathy through robust reconstruction of 3D retinal surfaces," *Computer Methods and Programs in Biomedicine*, vol. 137, pp. 1–10, 2016.
- [11] S. Khalid, M. U. Akram, T. Hassan, A. Nasim, and A. Jameel, "Fully automated robust system to detect retinal edema, central serous chorioretinopathy, and age related macular degeneration from optical coherence tomography images," *BioMed Research International*, vol. 2017, pp. 1–15, Article ID 7148245, 2017.
- [12] X. Xu, K. Lee, L. Zhang, M. Sonka, and M. D. Abramoff, "Stratified sampling voxel classification for segmentation of intraretinal and subretinal fluid in longitudinal clinical OCT data," *IEEE Transactions on Medical Imaging*, vol. 34, no. 7, pp. 1616–1623, 2015.
- [13] Z. Chen, D. Li, H. Shen, H. Mo, Z. Zeng, and H. Wei, "Automated segmentation of fluid regions in optical coherence tomography B-scan images of age-related macular degeneration," *Optics and Laser Technology*, vol. 122, Article ID 105830, 2020.
- [14] A. Rashno, D. D. Koozekanani, P. M. Drayna et al., "Fully automated segmentation of fluid/cyst regions in optical coherence tomography images with diabetic macular edema using neutrosophic sets and graph algorithms," *IEEE*

Transactions on Biomedical Engineering, vol. 65, no. 5, pp. 989–1001, 2018.

- [15] K. Alsaih, M. Yusoff, T. Tang, I. Faye, and F. Mériaudeau, "Deep learning architectures analysis for age-related macular degeneration segmentation on optical coherence tomography scans," *Computer Methods and Programs in Biomedicine*, vol. 195, Article ID 105566, 2020.
- [16] D. Lu, M. Heisler, S. Lee et al., "Deep-learning based multiclass retinal fluid segmentation and detection in optical coherence tomography images using a fully convolutional neural network," *Medical Image Analysis*, vol. 54, pp. 100–110, 2019.
- [17] B. Hassan, S. Qin, R. Ahmed et al., "Deep learning based joint segmentation and characterization of multi-class retinal fluid lesions on OCT scans for clinical use in anti-VEGF therapy," *Computers in Biology and Medicine*, vol. 136, Article ID 104727, 2021.
- [18] J. Xu, J. Shen, Z. Yan, F. Zhou, C. Wan, and W. Yang, "An intelligent location method of key boundary points for assisting the diameter measurement of central serous chorioretinopathy lesion area," *Computers in Biology and Medicine*, vol. 147, Article ID 105730, 2022.
- [19] R. Caruana, "Multi-task learning," *Machine Learning*, vol. 28, no. 1, pp. 41–75, 1997.
- [20] S. C. Huang, T. H. Le, and D. W. Jaw, "DSNet: joint semantic learning for object detection in inclement weather conditions," *IEEE Transactions on Pattern Analysis and Machine Intelligence*, vol. 43, no. 8, pp. 2623–2633, 2021.
- [21] Y. Zheng, J. Fan, J. Zhang, and X. Gao, "Hierarchical learning of multi-task sparse metrics for large-scale image classification," *Pattern Recognition*, vol. 67, pp. 97–109, 2017.
- [22] D. Sakkos, E. S. L. Ho, and H. P. H. Shum, "Illuminationaware multi-task GANs for foreground segmentation," *IEEE Access*, vol. 7, pp. 10976–10986, 2019.
- [23] L. Ju, X. Wang, X. Zhao et al., "Synergic adversarial label learning for grading retinal diseases via knowledge distillation and multi-task learning," *IEEE Journal of Biomedical and Health Informatics*, vol. 25, no. 10, pp. 3709–3720, 2021.
- [24] M. Kim, E. J. Min, K. Liu et al., "Multi-task learning based structured sparse canonical correlation analysis for brain imaging genetics," *Medical Image Analysis*, vol. 76, Article ID 102297, 2022.
- [25] Á. S. Hervella, J. Rouco, J. Novo, and M. Ortega, "End-to-end multi-task learning for simultaneous optic disc and cup segmentation and glaucoma classification in eye fundus images," *Applied Soft Computing*, vol. 116, Article ID 108347, 2022.
- [26] R. Sutton and A. Barto, "Reinforcement learning: an introduction," *IEEE Transactions on Neural Networks*, vol. 9, no. 5, p. 1054, 1998.
- [27] R. Liao, S. Miao, P. Tournemire et al., "An artificial agent for robust image registration," in *Proceedings of the In the Proceedings of the 31th AAAI Conference on Artificial Intelligence*, pp. 4168–4175, San Francisco, CA, USA, February 2017.
- [28] J. Krebs, T. Mansi, H. Delingette et al., "Robust non-rigid registration through agent-based action learning," in *Proceedings of the In the 20th International MICCAI Conference*, pp. 344–352, Quebec, Canada, September 2017.
- [29] B. Xu, J. Liu, X. Hou et al., "Attention by selection: a deep selective attention approach to breast cancer classification," *IEEE Transactions on Medical Imaging*, vol. 39, no. 6, pp. 1930–1941, 2020.
- [30] Z. Li and Y. Xia, "Deep reinforcement learning for weaklysupervised lymph node segmentation in CT images," *IEEE Journal of Biomedical and Health Informatics*, vol. 25, no. 3, pp. 774–783, 2021.

- [31] A. Alansary, O. Oktay, Y. Li et al., "Evaluating reinforcement learning agents for anatomical landmark detection," *Medical Image Analysis*, vol. 53, pp. 156–164, 2019.
- [32] F. Ghesu, B. Georgescu, T. Mansi, D. Neumann, J. Hornegger, and D. Comaniciu, "An artificial agent for anatomical landmark detection in medical images," in *Proceedings of the In the 19th International MICCAI Conference*, pp. 229–237, Athens, Greece, October 2016.
- [33] S. Kundeti, D. Parmar, A. S. MV, and D. Gautam, "Landmark detection in 3D medical images using reinforcement learning," in *Proceedings of the In the IEEE International Conference on Cloud Computing in Emerging Markets*, pp. 42–46, Bengaluru, India, November 2020.
- [34] F. C. Ghesu, B. Georgescu, Y. Zheng et al., "Multi-Scale deep reinforcement learning for real-time 3D-landmark detection in CT scans," *IEEE Transactions on Pattern Analysis and Machine Intelligence*, vol. 41, no. 1, pp. 176–189, 2019.
- [35] K. Dabov, A. Foi, V. Katkovnik, and K. Egiazarian, "Image denoising by sparse 3-D transform-domain collaborative filtering," *IEEE Transactions on Image Processing*, vol. 16, no. 8, pp. 2080–2095, 2007.
- [36] J. Deng, J. Guo, Y. Zhou, J. Yu, I. Kotsia, and S. Zafeiriou, "RetinaFace: single-stage dense face localisation in the wild," 2019, https://arxiv.org/abs/1905.00641.
- [37] K. He, X. Zhang, S. Ren, and J. Sun, "Deep residual learning for image recognition," in *Proceedings of the In the 2016 IEEE Conference on Computer Vision and Pattern Recognition* (CVPR), pp. 770–778, Las Vegas, NV, USA, June 2016.
- [38] A. Howard, M. Zhu, B. Chen et al., "MobileNets: efficient convolutional neural networks for mobile vision applications," 2017, https://arxiv.org/abs/1704.04861.
- [39] M. Sandler, A. Howard, M. Zhu, A. Zhmoginov, and L. Chen, "MobileNetV2: inverted residuals and linear bottlenecks," in Proceedings of the In the 2018 IEEE/CVF Conference on Computer Vision and Pattern Recognition, pp. 4510–4520, Salt Lake City, UT, USA, June 2018.
- [40] T. Lin, P. Dollar, R. Girshick, K. He, B. Hariharan, and S. Belongie, "Feature pyramid networks for object detection," in *Proceedings of the In the 2017 IEEE Conference on Computer Vision and Pattern Recognition (CVPR)*, pp. 936–944, Honolulu, HI, USA, July 2017.
- [41] M. Najibi, P. Samangouei, R. Chellappa, and L. Davis, "SSH: Single stage headless face detector," in *Proceedings of the In the* 2017 IEEE International Conference on Computer Vision (ICCV), pp. 4885–4894, Venice, Italy, October 2017.
- [42] C. J. Watkins and P. Dayan, "Q-learning," *Machine Learning*, vol. 8, no. 3/4, pp. 279–292, 1992.
- [43] V. Mnih, K. Kavukcuoglu, D. Silver et al., "Human-level control through deep reinforcement learning," *Nature*, vol. 518, no. 7540, pp. 529–533, 2015.
- [44] H. Hasselt, "Double Q-learning," in Proceedings of the In the 23rd International Conference on Neural Information Processing Systems, pp. 2613–2621, Vancouver, Canada, October 2010.
- [45] H. Hasselt, A. Guez, and D. Silver, "Deep reinforcement learning with double Q-learning," in *Proceedings of the In the Thirtieth AAAI Con-Ference on Artificial Intelligence*, pp. 2094–2100, Phoenix, AZ, USA, March 2016.
- [46] Z. Wang, T. Schaul, M. Hessel, H. Hasselt, M. Lanctot, and N. Freitas, "Dueling network architectures for deep reinforcement learning," in *Proceedings of the In the 1n the 33rd International Conference on International Conference on Machine Learning*, New York, NY, USA, June 2016.